Author's response to reviews

Title: Combined immunohistochemistry of beta-catenin, cytokeratin 7, and cytokeratin 20 is useful in discriminating primary lung adenocarcinomas from metastatic colorectal cancer

Authors:

Satoshi Ikeda (sikeda@hiroshima-u.ac.jp)
Masahiko Fujimori (f-macha@rd6.so-net.ne.jp)
Satoshi Shibata (ronshiba@hiroshima-u.ac.jp)
Masazumi Okajima (mokajima@hiroshima-u.ac.jp)
Yasuyo Ishizaki (ishizaki@chugoku.fofuku.go.jp)
Takeshi Kurihara (kurihara-tak@muc.biglobe.ne.jp)
Yoshihiro Miyata (ymiyata@hiroshima-u.ac.jp)
Masahiko Iseki (masaiseki@msn.com)
Yosuke Shimizu (gy_shimz@yahoo.co.jp)
Noriaki Tokumoto (tokumoto@hiroshima-u.ac.jp)
Shinji Ozaki (s-ozaki@bf7.so-net.ne.jp)
Toshimasa Asahara (asahara@hiroshima-u.ac.jp)

Version: 7 Date: 16 January 2006

Author’s response to reviews:

Thank you for your comments.
In accordance with your comments, we have thoroughly revised the Reference section and Figure.